Author's response to reviews

Title: The association between delusional-like experiences, and tobacco, alcohol or cannabis use: a nationwide population-based survey.

Authors:

Sukanta Saha (sukanta_saha@qcmhr.uq.edu.au)
James G Scott (James_G_Scott@health.qld.gov.au)
Daniel Varghese (Daniel_Varghese@health.qld.gov.au)
Louisa Degenhardt (louisa@burnet.edu.au)
Tim Slade (tims@unsw.edu.au)
John J McGrath (John_mcgrath@qcmhr.uq.edu.au)

Version: 5 Date: 1 December 2011

Author's response to reviews:

Dear Colleague,

We have included the extra text about future studies (as requested). The sentence is pasted in below and highlighted in yellow in the revision.

Kind regards,

John McGrath

In addition, future studies will explore the association between DLE versus (a) interactions between different substances, and (b) substances use plus other risk factors (e.g. trauma, family history).